Author's response to reviews

Title: The Four-Dimensional Symptom Questionnaire (4DSQ): a validation study of a multidimensional self-report questionnaire to assess distress, depression, anxiety and somatization

Authors:

Berend Terluin (b.terluin@vumc.nl)
Harm WJ van Marwijk (hwj.vanmarwijk@vumc.nl)
Herman J Ader (hj.ader@keyaccess.nl)
Henrica CW de Vet (hcw.devet@vumc.nl)
Brenda WJH Penninx (brendap@ggzba.nl)
Marleen LM Hermens (mlm.hermens@tiscali.nl)
Christine A van Boeijen (c.van.boeijen@degelderseroos.nl)
Anton JLM van Balkom (tonb@ggzba.nl)
Jac JL van der Klink (j.vanderklink@nspoh.nl)
Wim AB Stalman (w.stalman@vumc.nl)

Version: 3 Date: 18 August 2006

Author's response to reviews:

We have adapted the manuscript according to the formatting suggestions outlined in your e-mail of August 15.
Appendix 1 has become Table 1; Appendix 2 has become Appendix 1.
We have included the Acknowledgements and listed all sources of funding.